Dear [Editor's Name],

We would like to express our sincere gratitude for the valuable feedback provided by you and the reviewers on our manuscript titled "*[Manuscript Title]*" (Manuscript ID: [ID Number]). We appreciate the opportunity to revise our manuscript based on these insightful comments.

Below we address each of the specific comments raised by the reviewers:

Reviewer #1 Comments:

Comment 1: [Reviewer's comment]

Response: We have addressed this by [your response detailing changes made].

Comment 2: [Reviewer's comment]

Response: In response, we have [your response detailing changes made].

Reviewer #2 Comments:

Comment 1: [Reviewer's comment]

Response: To address this, we [your response detailing changes made].

Comment 2: [Reviewer's comment]

Response: We have incorporated [your response detailing changes made].

We hope that these revisions meet the expectations of the reviewers and contribute positively to the manuscript. A detailed explanation of the changes made is included in the attached document.

Thank you for considering our revisions. We look forward to your favorable response.

Sincerely, [Your Name] [Your Institution] [Your Email]